Author’s response to reviews

Title: MGAT1 is a novel transcriptional target of Wnt/β-catenin signaling pathway

Authors:

Izzet Akiva (dolphinarium02@yahoo.com)
Necla Birgül Iyison (birgul@boun.edu.tr)

Version: 3 Date: 28 Oct 2017

Author’s response to reviews:

1. Please revise your current Availability of Data and Materials statement to reflect one of the formats indicated in our submission guidelines:
   https://www.biomedcentral.com/getpublished/editorial-policies#availability+of+data+and+materials
   
   For example, please clearly state who should be contacted if someone wants to request the data in this section.

   Availability of Data and Materials statement is updated according to your submission guidelines and contacting author information is added.

2. We noted that the legend of Figure 3 does not include the subsections A, B and C which are cited in your manuscript. Please ensure that all figures are cited correctly in the main text and that all figure legends correspond to the cited sections.

   Legends, Figures and main text were updated so that they correspond correctly to each other and subsections were included wherever it is needed.